Author's response to reviews

Title: Risk factors of functional disability among community-dwelling elderly people by household in Japan: a prospective cohort study

Authors:

Emiko Saito (saito@hs.tmu.ac.jp)
Shouzoh Ueki (sueki@hss.tbu.ac.jp)
Nobufumi Yasuda (yasudan@kochi-u.ac.jp)
Sachiko Yamazaki (syamazaki@bgu.ac.jp)
Seiji Yasumura (yasumura@fmu.ac.jp)

Version: 3 Date: 7 October 2013

Author's response to reviews:

We thank your Editorial Board members for giving useful comments. In response to the Editorial Board members' comments, we have revised the manuscript 1546350917105827.

Comments: Your manuscript has been assessed by one of our Editorial Board members. According to their advice, and before proceeding further with your submission, we would ask you to please clarify if age and sex are controlled in tables 2 and 3.

Responses:

1. Table 2
We analyzed using the cox regression approach adjusted for age and sex. We rewrote the data of table2.

2. Table 3
We analyzed all data adjusted age and sex. Consequently, there was no significant variable in the children-only household group. Other results were approximately similar to the analysis before the adjustment. We rewrote the data of table3.